Astronomy 2: Session 2000-2001

TUTORIAL EXERCISES

Please hand in written solutions to the questions detailed below.

Note that model answers will be issued on the day that the written work is requested, so late scripts will not normally be accepted.

PLEASE WRITE YOUR ANSWERS ON SEPARATE SHEETS, since they are marked in parallel by the course lecturers.

Week Questions

To be handed in To be returned

22 _{(27/4/01)} 1.6, 4.9 Friday week 23

(4/5/01)

Friday week 24

(11/5/01)

24 _{(11/5/01)}

1.8, 4.1, 4.3 Friday week 25

(18/5/01)

Friday week 26

(25/5/01)

Tutorial Model Answers

Tutorial Exercise 7 Model Answer